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EXPLANATION OF ACID/ALKALI SEWE,B—NUMBERS
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% All numbers were token from records department and/or operating personnel.

¥ Acid sources (HC! plant productions) do not include inventory change . The crifice measuring
Per/Tri reactor vent flows to the HCI absorber was not accurately accounting for all the weak
acid produced in the absorber due to the erroneous assumption that the reactors were operating
at a constant pressure. The vent flows have been corrected for varying reactor pressures by a
factor of 0.803. This correction produced an HCI TPD that confirmed the total HCI TPD

 being measured fromthe bottom of the Per/Tri HCI absorber. The agreement of the two HCI

measurements assures greater efficiency in accounting for weak acid produced and consumed.

Miscellaneous acid sewers was an adjustment stream to balance the alkali presently being
sewered,

Alkali to Sewer .,

Silica Pigments' effluent to the sewer was calculated to have 13 TPD HCI equivalent of alkali.
This was calculated using the alkali reported from the £4 overflow and scaling up to the total
sewer flowrate, This was checked using the material balance for the Pigments plont.
Aa, 0
(2.5-gr/hit) (V%5) (8.3452 Ib/gal) = 0.02085 Ib/gal
(lit) (1000 gc) _“ko/lit_

Assume total plant sewer= 1500 gpm ab production rate of 100 TPD,
/

(0.02085 Jg/lga/l) (1500 g‘al/m}é) (1440 min) (1 ton) =22.5TPD Na,O
D 2000 T¥.

4
Lm/ A022.5 TPD (36.47 TPD HCI) " (T mole/D Na,0) = 13 TPD HCI equivalent 2677740 Lov ISE
le/DHCT) * (61.98 TPD Na,O) ASTFP Sot 26

- iy (T mo

Caustic for pH control was taken from the Records Department, as was tank 17 & 18 equivalent
TPD.

Alkali in the incinerator secondary serubber effluent was based on @ 14.5 gpm flowrate of
cell liquor to the secondary scrubber.

NoHCO3 + HCl  ———  NeCl + H,Q + CO,

1350 Ib/hr NeHCO, (36.47 Ih/hr HCI) = 586,06 Ib/hr = 7,03 TPD HCI
(82,01 Ib/hr NaHCO,)
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018 Sewer NaQH - HC1 Balance

Based on Conditions Tuc. .sc!.o!‘ A{-l“‘vrn 0O\ ZZZ"ZSJ

Acid Sources to 018 Flow (gpm) HC1 (gpl) HCL (TPD)
#1 WTU SS Eff1. .- 400 3% R.64

#2 WTU SS Effl. 006 58 15, 80
#1 Inc. Prim. Eff1l. S50 8. b2 2. 05

#2 Inc. Prim. Eff]. - L}Ow\d -
#3 Inc. Prim. Eff1. Diverted to o010 o

e —— e,

#4 Inc. Prim. Eff1. 2 SO Vo) O
TOTAL 2 00 0O gpm gpl 2 6.5 9 TPDHCL

Plant C - Cls 4+ Caustic I B 2 JOUTTY
FLOW = |2 80 gpm, 2.3 gpl sefrlly@m==s  TP]) NaQH
171 Tpp #Ci

ACID HEADER TOTAL 5. 079"

p
269 TPDHCI - (]9, 2 TPD Nam,\lt).z—';i—o_—ym = [ ~d—Fd~ TPD Xs HC

in Acid H

CAUSTIC HEADER TOTAL K9 B TFE ey

Silica @ 2 4 &0 gpm + 2.% gpl Na,0 = 23 ¥, 0 L-TPD Na.0

2
VDCM Rx-Bottoms Total ’L&t—%
A '
- R0 gm @ 20,4 gpl NaOH = =2, & 7 TPD NaOH

338 TPOwey

\o

018 Total HCl Before COS 75 7
\-
7.9’ 85 HCl 1.12 HC)
N [ ]
- - . == ity - . i Hme———= T
4% TPD HCI (3%.0% TPD Na,0) (=~ o/)/ (_R3¢4% 7). TPD Nao

2 s NaQH

( ( r:??-zg
Acid Hdr.) (Caustic Hdr.) o VDCM) - P
! &lz/ 47" TPD

“y B
-58.9

FNote ! Thiss/he Lalanse Lovry Lrdt did 9/io/pr.
.]Zl:l wif be done d’d;/, onl/ gon Sey &/,
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4 " | PPG Industries, inc. -am PROCESS CALCULATIONS
: Industrial Chemical Division Chemicals m Job_&-9/4 By LT
_ Lake Charles, Louisiana Date s~ //2/£3 Page of
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